Author’s response to reviews

Title: Ursodeoxycholic acid for the Prevention of symptomatic Gallstone disease after baRiAtric surgery: study protocol for a ranDomizEd controlled trial (UPGRADE trial)

Authors:

Thomas Boerlage (thomas.boerlage@slz.nl)
Sylke Haal (sylke.haal@slz.nl)
Maurits de Brauw (Maurits.deBrauw@slz.nl)
Yair Acherman (Yair.acherman@slz.nl)
Sjoerd Bruin (Sjoerd.bruin@slz.nl)
Arnold van de Laar (Arnold.vandelaar@slz.nl)
Daan Moes (Daan.moes@slz.nl)
Bart van Wagensveld (B.vanwagensveld@olvg.nl)
Claire de Vries (c.e.e.devries@olvg.nl)
Ruben van Veen (r.n.vanveen@olvg.nl)
Ruben Schouten (R.schouten@mcgroep.com)
Marcel Dijkgraaf (M.g.dijkgraaf@amc.uva.nl)
Paul Fockens (P.fockens@amc.uva.nl)
Victor Gerdes (Victor.gerdes@slz.nl)
Rogier Voermans (R.p.voermans@amc.uva.nl)

Version: 2 Date: 01 Oct 2017

Author’s response to reviews:

Dear editor,

Thank you for your review of our manuscript. We revised our manuscript as listed below.

- Please include a section in your Methods listing inclusion and exclusion criteria.
The inclusion and exclusion criteria are now given point-by-point under a separate heading.

- Please remove the "trial status" section from the Declarations.

  This was done.

- When you upload your revised manuscript, please include only those supplementary files that you wish to be published. Please include in the manuscript a list of supplementary files.

  This was done.